WILEY

Colorectal polyp outcomes after participation in the seAFOod polyp prevention trial: Evidence of rebound elevated colorectal polyp risk after short-term aspirin use

Amy Downing¹ | Hayley Fenton¹ | Claire Nickerson² | Paul M. Loadman³ | Elizabeth A. Williams⁴ | Colin J. Rees⁵ | Louise C. Brown⁶ | Eva J. A. Morris⁷ | Mark A. Hull¹

¹Leeds Institute of Medical Research, University of Leeds, Leeds, UK

²Public Health Commissioning, NHS England, London, UK

³Institute of Cancer Therapeutics, University of Bradford, Bradford, UK

⁴Department of Oncology and Metabolism, University of Sheffield, Sheffield, UK

⁵Population Health Sciences Institute, Newcastle University, Newcastle upon Tyne, UK

⁶MRC Clinical Trials Unit at University College, London, UK

⁷Nuffield Department of Population Health, University of Oxford, Oxford, UK

Correspondence

Mark A. Hull, Division of Gastrointestinal and Surgical Sciences, Leeds Institute of Medical Research, St James's University Hospital, University of Leeds, Leeds LS9 7TF, UK.

Email: m.a.hull@leeds.ac.uk

Funding information

Efficacy and Mechanism Evaluation Programme, Grant/Award Number: NIHR128210

Summary

Background: The seAFOod polyp prevention trial was a randomised, placebocontrolled, 2×2 factorial trial of aspirin 300mg and eicosapentaenoic acid (EPA) 2000mg daily in individuals who had a screening colonoscopy in the English Bowel Cancer Screening Programme (BCSP). Aspirin treatment was associated with a 20% reduction in colorectal polyp number at BCSP surveillance colonoscopy 12 months later. It is unclear what happens to colorectal polyp risk after short-term aspirin use. **Aim:** To investigate colorectal polyp risk according to the original trial treatment allocation, up to 6 years after trial participation.

Methods: All seAFOod trial participants were scheduled for further BCSP surveillance and provided informed consent for the collection of colonoscopy outcomes. We linked BCSP colonoscopy data to trial outcomes data.

Results: In total, 507 individuals underwent one or more colonoscopies after trial participation. Individuals grouped by treatment allocation were well matched for clinical characteristics, follow-up duration and number of surveillance colonoscopies. The polyp detection rate (PDR; the number of individuals who had \geq 1 colorectal polyp detected) after randomization to placebo aspirin was 71.1%. The PDR was 80.1% for individuals who had received aspirin (odds ratio [OR] 1.13 [95% confidence interval 1.02, 1.24]; p = 0.02). There was no difference in colorectal polyp outcomes between individuals who had been allocated to EPA compared with its placebo (OR for PDR 1.00 [0.91, 1.10]; p = 0.92).

Conclusion: Individuals who received aspirin in the seAFOod trial demonstrated increased colorectal polyp risk during post-trial surveillance. Rebound elevated ne-oplastic risk after short-term aspirin use has important implications for aspirin cessation driven by age-related bleeding risk. ISRCTN05926847.

The Handling Editor for this article was Dr Colin Howden, and it was accepted for publication after full peer-review.

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2023 The Authors. Alimentary Pharmacology & Therapeutics published by John Wiley & Sons Ltd.

1 | INTRODUCTION

The seAFOod polyp prevention trial was a randomised, double-blind, placebo-controlled, 2×2 factorial trial of the colorectal cancer (CRC) chemoprevention efficacy of aspirin 300 mg daily and eicosapentaenoic acid (EPA) 2000 mg free fatty acid equivalents daily in 'high risk' patients undergoing colonoscopy surveillance in the English Bowel Cancer Screening Programme (BCSP).^{1,2} Trial participants were aged 55-73 years and had been invited for screening colonoscopy on the basis of a positive faecal occult blood test or 'high risk' screening flexible sigmoidoscopy.^{1,2} The trial population was predominantly (80%) male and White European, in keeping with the demographic characteristics of individuals undergoing BCSP colonoscopy.^{1,2} Patients with a known genetic CRC predisposition were excluded.^{1,2} The intervention period between the screening colonoscopy and the first surveillance colonoscopy was 12 months in individuals deemed 'high risk' (defined as ≥ 5 polyps or ≥ 3 polyps, if one or more polyps were ≥10mm in size), a duration which has been associated with a similar degree of colorectal polyp risk reduction compared with a 3year intervention period in previous aspirin polyp prevention trials.³

The primary finding from the seAFOod trial was that aspirin and EPA did not reduce colorectal polyp incidence, measured as the 'adenoma detection rate' (the % of individuals with one or more colorectal polyps 12 months after clearance colonoscopy) by an 'at the margins' analysis of the individual interventions.^{1,2} However, aspirin use was associated with a significant reduction in overall colorectal polyp risk (measured as mean polyp number per participant).^{1,2} There was also colorectal site- and polyp type (conventional adenoma or serrated polyp)-specific chemoprevention activity of aspirin and EPA; notably, randomisation to aspirin was associated with reduced risk of serrated lesions, unlike EPA treatment, which was associated with a statistically significant reduction in risk of left-sided (distal to the splenic flexure) conventional adenomas.^{1,2} The seAFOod trial was not powered for a pre-specified 'inside the table' analysis of the four treatment groups, including combined aspirin and EPA therapy.^{1,2} However, laboratory studies have since suggested that aspirin and EPA may have a positive interaction for CRC risk prevention via synthesis of novel oxylipins such as E-type resolvins,⁴ and/or substrate diversion of EPA secondary to cyclooxygenase inhibition by aspirin.⁵

It is not known whether CRC chemoprevention agents inhibit either, or both, initiation and/or growth of tumours at the earliest stages of intestinal tumorigenesis. Inhibition of tumour initiation is hypothesised to provide prolonged benefit after cessation of a chemoprevention agent. Alternatively, tumour growth suppression alone (without inhibition of tumour initiation) is hypothesised to lead to a 'rebound' increase in colorectal polyp incidence, whereby undetectable tumours that initiated but did not grow to become macroscopically visible during chemopreventative agent use, are de-repressed and become detectable upon cessation of chemoprevention therapy. Knowledge of whether a 'rebound' increase in colorectal polyps (indicative of increased CRC risk) occurs will be critical in order to define guidelines for the duration of chemoprevention use, optimal timing of cessation of therapy and best use of

 $\operatorname{AP}_{\!\!\!\&}\mathrm{T}$ Alimentary Pharmacology & Therapeutics $-\mathbf{W}$ I LEY

accompanying colonoscopic surveillance. This is particularly important in the elderly, in whom the common practice is to stop aspirin therapy driven by increasing concern about elevated bleeding risk.⁶

seAFOod trial participants were invited by the English BCSP to undergo surveillance colonoscopy at 3 years (or earlier dictated by prior colorectal polyp findings) after the initial 'high risk' 1-year surveillance procedure (which was the seAFOod trial exit colonoscopy), with subsequent surveillance dependent on the most recent findings (Figure S1). All trial participants provided informed consent for collection and use of post-trial BCSP data up to 6 years after trial participation (covering a maximum of two 3-year surveillance cycles; Figure S1). Therefore, we obtained post-trial colonoscopy data in order to investigate colorectal polyp risk following cessation of short-term chemoprevention with aspirin and EPA.

2 | METHODS

2.1 | Bowel cancer screening programme data

English BCSP Screening System (BCSS) data were provided under approvals from the Public Health England Office for Data Release (ODR1920_199 dated 6/9/2021) and the BCSP Research Advisory Committee (BCSPRAC_285). This project was a component of the STOP-ADENOMA study, which had approval from the London and Surrey Borders Research Ethics Committee (19/LO/1655) and is registered as ISRCTN05926847.

All participants in the seAFOod polyp prevention trial provided specific, written informed consent to allow access to post-trial BCSS data.^{1,2} Endoscopy surveillance data on procedures that occurred in the BCSP for a maximum of 6 years after the seAFOod trial exit colonoscopy (which was the first 'high risk' BCSP surveillance colonoscopy at 12 months after screening) were obtained, where linkage was possible (Figure S1). The first trial exit colonoscopy was in November 2012 and the last trial exit colonoscopy was in June 2017.

The BSCP data extraction date was 7 October 2021. BCSS data were linked to the seAFOod trial database using date of birth, sex and hospital site, as well as dates of the screening and first surveillance colonoscopy. No other data on seAFOod trial participants were available after the trial finished, including subsequent use of aspirin and/or omega-3 polyunsaturated fatty acid supplements. Participants were not told their treatment allocation after the trial had completed and the Plain English Summary of the trial described the null primary outcome of the trial without any recommendation for future CRC chemoprevention.

2.2 | Data collection

Data were obtained at individual-, test (each endoscopic procedure)-, BCSP episode (which can include one or more endoscopic procedures as a single administrative clinical episode)- and polyp-level. For each post-trial surveillance colonoscopy, the time elapsed since the first

DOWNING ET AL.

surveillance colonoscopy and extent of examination (complete or incomplete colonoscopy) were noted, as well as the number of separate procedures per individual. Endoscopic procedures occurring within 6 months of the date of the trial exit colonoscopy were assumed to be part of the same episode of BCSP care as the surveillance colonoscopy at 12 months after screening and were discounted.

Colorectal cancer detection during BCSP colonoscopic surveillance was stipulated as a key outcome. Colorectal polyp analysis was stratified according to total number of polyps, polyp type (adenoma or serrated-hyperplastic), polyp size (histological size as per BCSP guidelines), polyp location (proximal [right-sided]-proximal to splenic flexure; distal [left-sided]-at and distal to splenic flexure), and on the basis of 'advanced' features (defined as ≥10mm by histology size, and/or high-grade dysplasia and/or any villous architectural component), in keeping with criteria used in the seAFOod trial.^{1,2} Although the terminology and classification of serrated and hyperplastic polyps have changed since the start of the seAFOod trial in 2011, with the acceptance that diminutive (≤ 5 mm) rectal hyperplastic polyps can be ignored for the purposes of risk stratification for future surveillance,⁷ we continued to report a combined serratedhyperplastic polyp category in order to facilitate comparison with the original seAFOod trial results.^{1,2}

Colorectal polyp incidence was reported as the 'polyp detection rate' (PDR; calculated as the % of individuals with one or more colorectal polyps detected during the total surveillance period). The colorectal polyp number, size in millimetres (the mean value of the maximum dimension of all polyps detected per individual, measured histologically [or endoscopically if the complete polyp could not be assessed]) and 'burden' (defined as the sum of the individual polyp diameters) were reported as the mean value per person (for example, mean polyp number per person [MPP]).

In addition, the BCSP classification of colonoscopic outcome (normal, abnormal [but no polyp], one or more polyps [sub-classified as 'high-risk' {≥5 polyps or ≥3 polyps if one or more are ≥10mm in size}, 'intermediate-risk' {3-4 subcentimetre polyps}, 'low-risk' {1-2 subcentimetre polyps]]), which was in use during the seAFOod trial, was used to report findings at procedural and BCSP episode level.⁸ Since the seAFOod trial finished, updated British Society of Gastroenterology (BSG) Guidelines on colonoscopic surveillance after polypectomy have been adopted by the BSCP since mid-2020.⁷ Current risk stratification now consists of a single 'high-risk' category (≥5 polyps or ≥2 polyps if one or more are ≥10mm in size) for 3-year surveillance, as opposed to other findings that do not warrant further colonoscopic surveillance. Therefore, data were also analysed at test and episode level according to the new BCSP classification, to ensure that the results reflected current BCSP practice and current US Multi-Society Task Force on CRC guidelines on colonoscopic surveillance.⁹

The baseline characteristics (age, sex, body mass index [BMI], smoking status) of seAFOod trial participants that had post-trial BCSP surveillance data (included in this analysis) versus trial participants that did not have any linked endoscopic BCSP surveillance data after the seAFOod trial were compared.

2.3 | Statistical analysis

2.3.1 | Post-trial surveillance colonoscopy outcomes

The primary outcome for the post-trial analysis was stipulated as total colorectal polyps by an 'at the margins' analysis, consistent with the primary analysis of the seAFOod trial,^{1,2} by comparing data from those randomised to aspirin versus no aspirin, and individuals randomised to EPA versus no EPA.

Colorectal polyp data are reported as the % value for each PDR outcome, as well as the mean and 95% confidence interval (CI) values for colorectal polyp number (MPP). Secondary colorectal polyp size and burden data are also reported as mean and 95% CI values. The Student's t-test or one-way analysis of variance was used for univariate comparison of continuous participant characteristics and follow-up duration across two or more groups, as appropriate. A chi-squared test was used to compare sex ratios across treatment groups. Treatment group differences in PDR were tested by logistic regression. Data are reported as the odds ratio (OR) and 95% CI. A negative-binomial regression model was used to compare colorectal polyp number (and polyp burden) outcomes based on the skewed distribution of individual colorectal polyp frequencies that was apparent at the trial exit colonoscopy (Figure S2).^{1,2} The incidence rate ratio (IRR) and 95% CI values for groups based on prior seAFOod trial treatment allocation are reported for comparison with the respective placebo group. The mean colorectal polyp size per person was compared across previous trial treatment allocation by linear regression and reported as the mean size difference and 95% CI. All models were adjusted for age and sex and also included BCSP research site as a random effect. Statistical significance for all analyses was assumed at *p* ≤ 0.05.

In addition to the main analysis of cumulative post-trial colorectal polyp outcomes, which included BCSP surveillance data from colonoscopies for up to 6 years after the trial exit colonoscopy (Figure S1), we also performed a sensitivity analysis of total colorectal polyp outcomes at a single point in time after seAFOod trial participation. This was restricted to individuals who had 3-year surveillance colonoscopy outcomes only (using a 2.5- to 3.5-year post-trial window), excluding individuals who had undergone any post-trial procedure prior to this.

2.3.2 | seAFOod trial colonoscopy outcomes

We repeated the 'at the margins' analysis of total colorectal polyp number from the seAFOod trial in the subgroup of trial participants that had post-trial colonoscopy data, in order to confirm the original trial treatment effect of aspirin on total colorectal polyp risk in the current study cohort. The same Poisson regression model included in the original final trial analysis, adjusted for the BCSP research site as a random effect and for repeat colonoscopy at trial baseline, was used.^{1.2} A separate negative-binomial model ${
m AP}_{\!\&}{
m T}$ Alimentary Pharmacology & Therapeutics – WILEY

was also used based on the distribution of individual colorectal polyp frequencies that was apparent at the trial exit colonoscopy (Figure S2).^{1,2}

In addition, we performed post hoc 'inside the table' analysis of seAFOod trial colorectal polyp outcomes, comparing the four individual treatment groups (placebo aspirin and placebo EPA [placebo]; active aspirin and placebo EPA [aspirin]; placebo aspirin and active EPA [EPA]; active aspirin and active EPA [aspirin and EPA]), on the basis that the total number of colorectal polyps detected at the seAFOod trial exit colonoscopy in the group receiving both aspirin and EPA together was noticeably less than in the groups who received either agent alone, or placebo only.^{1,2} Using the original seAFOod trial outcomes data and the same regression model used in the primary trial analysis, we generated the IRR and 95% CI for the group that received combined treatment compared with the other three groups.^{1,2} An interaction between aspirin and EPA was analysed by comparing individuals allocated to active EPA with those who received placebo EPA, using a Poisson regression model,^{1,2} which also included age, sex, BMI, smoking status, alcohol intake, baseline red blood cell % EPA level and randomisation to aspirin as co-variables, with statistical significance reported as the Wald test for interaction. Evidence of an interaction between aspirin and EPA prompted a secondary comparison of the posttrial surveillance colonoscopy outcomes across the four treatment groups, in addition to the primary analysis according to factorial margins.

2.4 | Patient and public involvement

A patient and public representative was a member of the STOP-ADENOMA study group and supported the application to the Public Health England Office for Data Release for access to the BCSP data, by writing a General Data Protection Regulation Privacy Notice in conjunction with the co-investigators.

3 | RESULTS

3.1 | The post-trial study population and BCSP procedures

Five hundred and seven individuals, who had been randomised to the seAFOod trial, had undergone one or more colonoscopies in the English BCSP, which were more than 6 months after, and less than 6 years after, trial participation (Figure 1A). No BCSS data were available for 200 participants for several reasons (Figure 1A). Only 71 individuals were invited for BCSP colonoscopy but did not attend, a number which is likely to have been increased by the ongoing COVID-19 pandemic (Figure 1A).

The 507 individuals, for whom post-trial surveillance colonoscopy data were available, were distributed equally across the four seAFOod trial treatment groups (placebo 127 [72.2% of the original trial treatment group]; aspirin 128 [72.7%]; EPA 129 [72.5%]; aspirin + EPA 123 [69.5%]) and displayed similar demographic and clinical characteristics to the overall seAFOod trial population, which was predominantly male with excess body weight (Table S1).^{1,2}

We confirmed that the post-trial surveillance study population of 507 individuals displayed a similar on-trial treatment effect size for aspirin and EPA as the published primary analysis of the full seAFOod trial population.^{1,2} The IRR (95% CI) for aspirin treatment versus no aspirin was 0.77 (0.66, 0.91), and 0.93 (0.79, 1.09) for EPA versus no EPA, for total colorectal polyps, thereby confirming the chemoprevention activity of aspirin in the seAFOod trial sub-population, for which post-trial data were available. The corresponding IRR (95% CI) values from the negative binomial model were 0.78 (0.61, 0.99) and 0.91 (0.71, 1.17) for aspirin and EPA, respectively.

The study population underwent 602 colonoscopies in 574 separate BCSP episodes with the majority (76.6%) of procedures corresponding to the expected 3-year surveillance colonoscopy window (2.5–3.5 years after the first surveillance [trial exit] colonoscopy) for individuals without 'high risk' findings (Figure 1B and Figure S1). Four hundred and thirty (84.8%) individuals in the study population underwent only one colonoscopy during the follow-up period. Ninetysix per cent (n=551) of BCSP episodes during the study period consisted of only one colonoscopy (Figure S3).

Colonoscopy was reported as complete with insertion to the caecum in 588 (97.7%) of 602 procedures (Figure S4). Diagnostic outcomes according to the BCSP classification recorded in the BCSS at individual colonoscopy and episode level are reported in Figure S4. We also re-classified surveillance outcomes according to the current risk stratification used by the English BCSP surveillance pathway.⁸ Overall, 96 (16.0%) of 602 procedures were classified as 'high-risk' (\geq 5 polyps, or \geq 2 polyps if one or more are \geq 10mm in size) and 359 (59.6%) procedures found at least one colorectal polyp but did not fulfil current 'high-risk' criteria. Findings were similar at episode level with 90 (15.7%) of 574 BCSP episodes that occurred during the posttrial follow-up period being classified as 'high risk'.

In total, 1298 'polyps' were recorded, of which 936 (72.1%) were adenomas and 273 (21.0%) were serrated-hyperplastic polyps (Table 1). Current BCSP guidance is clear that diminutive (≤5 mm) rectal, hyperplastic-looking polyps are not recorded,⁷ but practice was more variable during the seAFOod trial and the follow-up period encompassed by this study. Therefore, a small number (n=15)of diminutive rectal hyperplastic polyps were recorded in the BCSS and are included in the analysis (Table 1). Overall, 750 (57.8%) polyps were located proximal to the splenic flexure as opposed to 548 (42.2%) polyps at or distal to the splenic flexure. There were 77 'advanced' adenomatous polyps in 54 individuals, 40 of which were ≥10mm in size. Thirty-eight 'advanced' polyps exhibited ≥25% villous architecture and three lesions displayed high-grade dysplasia. No CRCs were detected during post-trial colonoscopic surveillance of the 507 individuals who were originally deemed 'high risk' at screening colonoscopy.

565







FIGURE 1 Post-trial colonoscopy surveillance of seAFOod trial participants. (A) Linkage of the 707 seAFOod trial participants who provided trial data to post-trial BCSP colonoscopy surveillance episodes. The reasons why no BCSS data were available for 200 participants are listed. Ninety-seven participants were not invited for surveillance colonoscopy as per BCSP guidelines. (B) Duration between the seAFOod trial exit colonoscopy (first surveillance colonoscopy) and subsequent post-trial surveillance colonoscopies (Figure S1). BCSP, Bowel Cancer Screening Programme; BCSS, Bowel Cancer Screening System.

3.2 | The effect of seAFOod trial interventions on post-trial colorectal polyp outcomes

Respective active versus placebo intervention groups were wellmatched for the duration of post-trial follow-up and the number of surveillance procedures that had been performed per person (Table 2). The PDR after being randomised to placebo aspirin was 71.1%. By contrast, the PDR was 80.1% for individuals who had received active aspirin for 1 year during the seAFOod trial (OR 1.13 [1.02, 1.24]; p = 0.02). A similar increase in PDR in the group that had received aspirin, as opposed to its placebo, was observed for conventional adenomas (69.3% vs. 59.4%; OR 1.16 [1.03, 1.32]; p = 0.02), but not for serrated-hyperplastic polyps (31.5% vs. 27.7%; OR 1.10 [0.84, 1.44]; p = 0.47).

In addition, the number of colorectal polyps detected during colonoscopic surveillance was higher in those who had received aspirin during trial participation (MPP 2.7) compared with the group who received placebo aspirin (MPP 2.5), but the difference was not statistically significant (IRR for total colorectal polyps 1.11 [0.90, 1,36]; p=0.32: Table 2).

By contrast, prior aspirin users did not demonstrate the altered risk of advanced colorectal polyps during post-trial follow-up compared with individuals previously allocated to placebo aspirin (IRR 1.04 [0.63, 1.71]; Table 2).

There was no difference in PDR (OR for total colorectal polyps 1.00 [0.91, 1.10]; p=0.92) or the number of colorectal polyps (IRR 1.10 [0.90, 1.36]; p=0.35) detected during post-trial BSCP surveillance between individuals, who had received either active or placebo EPA during the seAFOod trial (Table 2).

Secondary analysis of colorectal polyp size demonstrated that individuals, who were allocated to aspirin treatment during the seAFOod trial, also had larger colorectal polyps detected during post-trial BCSP surveillance than those who received placebo aspirin (Table S2). This was evident for total colorectal polyps (size difference + 0.48 mm [+0.09, +0.86]; p=0.02) and was most marked for serrated-hyperplastic polyps (size difference + 1.09 mm [+0.29, +1.89]; p=0.01). However, no significant difference was observed in colorectal polyp size, who did or did not previously receive EPA (size difference for total colorectal polyps +0.10 mm [-0.28, 0.49]; p=0.60; Table S2). TABLE 1 Histological type and location of colorectal polyps removed during colonoscopic surveillance after seAFOod trial participation.

Polyp histology		Polyp location			
	n	%		n	%
Adenoma	936	72.1	Rectum	117	9.0
Serrated-hyperplastic	273	21.0	Sigmoid colon	225	17.3
Serrated ^a	40		Descending colon	119	9.2
Hyperplastic ≥10mm	9		Splenic flexure	87	6.7
Hyperplastic <10 mm ^b	224		Transverse colon	299	23.0
Inflammatory polyp	13	1.0	Hepatic flexure	50	3.9
Lymphoid aggregate	4	0.3	Ascending colon	269	20.7
Histology not available	72	5.5	Caecum	132	10.2
Total	1298	100	Total	1298	100

AP&T Alimentary Pharmacology & Therapeutics – WILEY

^a36 sessile serrated lesions, one sessile serrated adenoma and three traditional serrated adenomas. ^b15 rectal hyperplastic polyps \leq 5 mm.

We also calculated the cumulative colorectal polyp burden, combining colorectal polyp number and size as a readout of overall neoplastic recurrence (Table S2). Consistent with the colorectal polyp number and size data, the total colorectal polyp and serrated-hyperplastic polyp burden in individuals, who had been randomised to aspirin during the seAFOod trial, was larger than those who had received placebo aspirin, with statistical significance for the increase in serrated-hyperplastic polyp burden (IRR 1.28 [1.04, 1.59]; p=0.02; Table S2).

Although the groups that were defined by the prior trial intervention were well-matched for clinical characteristics, length of post-trial surveillance and the number of surveillance procedures per individual (Table 2), the complex relationship between colorectal polyp detection and subsequent BCSP risk stratification, thus driving the number and timing of colonoscopies, and hence chance of colorectal polyp detection, could potentially introduce bias. Therefore, we also performed a sensitivity analysis of total colorectal polyp outcomes that was restricted to the three-year surveillance colonoscopy. In 444 seAFOod trial participants, who had 3-year surveillance colonoscopy data only (in which 885 colorectal polyps were reported), total colorectal polyp analysis confirmed the findings of the main analysis (Figure S2 and Table S3). There was a higher PDR (75.0%) in individuals that had been allocated to active aspirin in the seAFOod trial compared with those allocated to placebo aspirin (PDR 67.7%, or 1.10 [0.98, 1.24], which just failed to reach statistical significance; p=0.11; Table S3). However, the number of total colorectal polyps was higher in individuals previously allocated to aspirin as opposed to placebo aspirin (IRR 1.25 [1.00, 1.58]; p=0.05; Table S3).

3.3 | Evidence for an interaction between aspirin and EPA for reduction in total colorectal polyp risk

In addition to the pre-specified analysis of colorectal polyp outcomes in the seAFOod trial, we noted that the number of colorectal polyps detected in individuals who had been randomised to both aspirin and EPA was lower than the other three treatment groups.^{1,2} Using data from all seAFOod trial participants (n=707), direct comparison of the group that received combined treatment with the groups that received either a single agent or placebo only demonstrated that there was a significantly lower risk of any colorectal polyp (25%-35%) in individuals who had received both aspirin and EPA compared with the other trial groups (Table 3). Further analysis stratified for randomisation to active EPA or placebo EPA, demonstrated a significant interaction with allocation to active aspirin compared with placebo aspirin (IRR for EPA vs no EPA 0.60 [0.43–0.85] in aspirin users compared with 1.04 [0.75–1.44] in those receiving placebo aspirin [p_{int} =0.01]; Table 4).

In view of the unexpected interaction between aspirin and EPA for the reduction in colorectal polyp risk, analysis of the post-trial colonoscopy data was also performed 'inside the table' across all four treatment groups (Table S4).^{1,2} Results mirrored the 'at the margins' analysis of the individual interventions shown in Table 2 with an increase in PDR for total colorectal polyps and distal colorectal polyps in the treatment combinations that included active aspirin compared with the group that received placebo aspirin and placebo EPA only (Table S4). A similar relationship with previous single-agent aspirin use was also observed for increased distal colorectal polyp number (Table S4). Prior use of EPA alone was associated with an increased PDR for distal colorectal polyps (OR 1.31 [1.00, 1.70]; p = 0.05), reflecting a reciprocal relationship with the decrease in risk of distal colorectal polyps linked to EPA use during the seAFOod trial.^{1,2} Previous treatment with EPA alone was also associated with increased serrated-hyperplastic polyp number during follow-up (IRR 1.62 [1.05, 2.50]; p=0.03). Furthermore, analysis of total colorectal polyp size demonstrated that those individuals, who had been randomised to a combination of aspirin and EPA treatment during the seAFOod trial, had larger colorectal polyps (mean size difference + 0.56 [+0.02, +1.11]; p=0.04) that was explained by the increased size of serrated-hyperplastic polyps (mean size

567

Group characteristics	No aspirin (n=256)	Aspirin (<i>n</i> = 251)	р	No EPA (n=255)	EPA (n = 252)	p
Age (years) ^a	65 (64, 65)	64 (64, 65)	0.76	64 (64, 65)	64 (64, 65)	0.58
Sex (male [%]:female)	200 (78):56	204 (81):47	0.38	202 (79):53	202 (80):50	0.79
Duration of post-trial follow-up (days) ^a	1188 (1160, 1217)	1218 (1186, 1250)	0.17	1216 (1186, 1246)	1189 (1158, 1220)	0.22
Number of colonoscopies per person ^a	1.2 (1.1, 1.3)	1.2 (1.1, 1.3)	0.86	1.2 (1.1, 1.3)	1.2 (1.1, 1.2)	0.35
PDR	% (no. with ≥1 polyp)	% (no. with ≥1 polyp)	OR (95% CI) <i>p</i> value ^b	% (no. with ≥1 polyp)	% (no. with ≥1 polyp)	OR (95% CI) <i>p</i> value ^b
Total colorectal polyp	71.1 (182)	80.1 (201)	1.13 (1.02, 1.24) $p = 0.02$	75.3 (192)	75.8 (191)	1.00 (0.91, 1.10) p=0.92
Advanced polyp	10.5 (27)	10.8 (27)	1.04 (0.63, 1.71) <i>p</i> =0.89	9.8 (25)	11.5 (29)	1.16 (0.70, 1.92) <i>p</i> =0.57
Adenoma	59.4 (152)	69.3 (174)	1.16 (1.03, 1.32) $p = 0.02$	65.5 (167)	63.1 (159)	0.96 (0.85, 1.09) p=0.53
Serrated-hyperplastic	27.7 (71)	31.5 (79)	1.10 (0.84, 1.44) <i>p</i> =0.47	27.1 (69)	32.1 (81)	1.19 (0.91, 1.55) <i>p</i> =0.21
Distal polyp	46.5 (119)	54.2 (136)	1.15 (0.97, 1.37) <i>p</i> =0.11	48.6 (124)	52.0 (131)	1.04 (0.88, 1.24) <i>p</i> =0.63
Proximal polyp	57.8 (148)	60.2 (151)	1.03 (0.89, 1.19) <i>p</i> =0.67	59.2 (151)	58.7 (148)	1.00 (0.87, 1.15) <i>p</i> = 1.00
Colorectal polyp number	MPP (mean [95% CI])	MPP (mean [95% Cl])	IRR (95% CI) <i>p</i> value ^c	MPP (mean [95% CI])	MPP (mean [95% Cl])	IRR (95% CI) <i>p</i> value ^c
Total colorectal polyp	2.5 (1.9, 3.0)	2.7 (2.3, 3.1)	1.11 (0.90, 1.36) $p = 0.32$	2.4 (2.0, 2.9)	2.7 (2.2, 3.2)	1.10 (0.90, 1.36) <i>p</i> =0.35
Advanced polyp	0.17 (0.10, 0.25)	0.13 (0.08-0.19)	0.77 (0.47, 1.25) <i>p</i> =0.29	0.14 (0.07, 0.20)	0.17 (0.10, 0.23)	1.21 (0.75, 1.96) <i>p</i> =0.47
Adenoma	1.8 (1.3, 2.2)	1.9 (1.6, 2.3)	1.15 (0.92, 1.43) p=0.22	1.8 (1.4, 2.2)	1.9 (1.5, 2.3)	1.05 (0.84, 1.30) $p = 0.84$
Serrated-hyperplastic	0.51 (0.34, 0.67)	0.57 (0.38, 0.76)	1.13 (0.84, 1.51) $p = 0.44$	0.49 (0.32, 0.66)	0.59 (0.41, 0.77)	1.20 (0.89, 1.62) <i>p</i> =0.23
Distal polyp	1.0 (0.8, 1.2)	1.1 (0.9, 1.4)	1.13 (0.89, 1.44) <i>p</i> =0.32	1.0 (0.8, 1.2)	1.2 (0.9, 1.4)	1.18 (0.92, 1.50) <i>p</i> =0.19
Proximal polyp	1.4 (1.0, 1.8)	1.5 (1.3, 1.8)	1.08(0.86, 1.36)p = 0.50	1.4 (1.1, 1.8)	1.5 (1.2, 1.9)	1.05(0.84, 1.32)p=0.54

TABLE 2 Comparison of colorectal polyp outcomes during post-trial colonoscopic surveillance between individuals who received either active intervention or its respective placebo in the 2×2 factorial seAFOod trial.

Note: Data per person are cumulative counts during the total surveillance period for each individual.

Abbreviations: IRR, incidence risk ratio; MPP, mean polyp number per person; OR, odds ratio; PDR, polyp detection rate.

^aData are expressed as the mean value and 95% confidence interval (CI).

^bStatistical significance between groups originally allocated placebo or active interventions was tested by logistic regression adjusted for age and sex. ^cStatistical significance between groups originally allocated placebo or active interventions was tested by negative binomial regression adjusted for age and sex. Statistically significant findings are highlighted in bold text.

difference + 1.44 [+0.29, +2.60]; p = 0.01) during post-trial surveillance (Table S2). Use of the colorectal polyp burden endpoint amplified the relationship between prior EPA or aspirin use and increased serrated-hyperplastic polyp recurrence (Table S2).

'Inside the table' analysis of the post-trial colonoscopy outcomes restricted to 3-year procedures confirmed that the increased PDR compared with previous placebo users was explained by the group that had received both aspirin and EPA 3 years' earlier (OR 1.20 [1.01, 1.42]; p=0.04), rather than the those who received either agent alone (aspirin only OR 1.15 [0.96, 1.37] and EPA only OR 1.12 [0.94, 1.35]; both p>0.1). The total colorectal polyp number per person in those previously allocated to combination of aspirin and EPA was also higher than those that had been allocated to placebo (MPP 2.28 [1.82, 2.74] vs. 1.68 [1.18, 2.17]; IRR 1.40 [1.01, 1.93], p=0.04), unlike the groups that had received aspirin (1.22 [0.88, 1.70]; p=0.24), or EPA (1.09 [0.78, 1.52]; p=0.63), alone, compared with those who received placebos only.

4 | DISCUSSION

This pre-specified analysis of colorectal polyp risk during routine colonoscopic surveillance in the English BCSP, following participation in the seAFOod polyp prevention trial, has demonstrated that those individuals who originally received aspirin in the seAFOod trial (which was associated with *decreased* colorectal polyp risk as measured by colorectal polyp number^{1,2}) actually had *increased* cumulative (total and adenomatous) colorectal polyp risk during post-trial surveillance compared with those previously allocated to placebo aspirin.

These data support the hypothesis that short-term aspirin use inhibits colorectal tumour growth, but not initiation, leading to derepression of polyp growth upon treatment cessation, reflected as 'rebound' increased colorectal polyp risk during subsequent colonoscopic surveillance. This relationship was restricted to aspirin, which had chemopreventative efficacy against adenomatous and serrated TABLE 3 Comparison of total colorectal polyp number in individuals who received combined aspirin and EPA treatment in the seAFOod trial with participants who received either agent alone or placebos only.

	Incidence risk ratio (95% CI) for the active aspirin and active EPA group ^a
Placebos only	0.70 (0.57, 0.86)
Placebo aspirin and active EPA	0.65 (0.53, 0.80)
Active aspirin and placebo EPA	0.75 (0.61, 0.93)

^aAdjusted by BCSP site as a random effect and for repeat colonoscopy at baseline in the seAFOod trial [see ref. 1, 2].

polyps in the seAFOod trial (which was also confirmed in the posttrial surveillance population),^{1,2} whereas allocation to active EPA in the seAFOod trial (which was not linked to reduced risk of total colorectal polyps) was not associated with altered colorectal polyp risk during post-trial follow-up, according to factorial trial margins. 'Inside the table' treatment group analysis of colorectal polyp findings during post-trial surveillance also reflected individual agent chemopreventative efficacy during the seAFOod trial, with prior EPA use, which was associated with reduced on-trial distal colorectal polyp risk,^{1,2} being associated with increased distal colorectal polyp risk during post-trial surveillance.

The effect size (a 10%–15% increase in PDR and MPP in those allocated aspirin, as opposed to placebo during the seAFOod trial) is clinically meaningful based on a similar effect size for colorectal polyp risk reduction in aspirin polyp prevention trials that is coupled with observational data on CRC risk reduction,^{3,10,11} as well as PDR differences reported for endoscopic interventions that improve colorectal polyp detection, which have been adopted into clinical practice.¹²

Secondary colorectal polyp size and polyp burden analyses concurred with the PDR outcomes in that trial allocation to aspirin was associated with larger colorectal polyps, including serrated lesions. In general, serrated-hyperplastic polyps detected during surveillance were larger than adenomatous polyps. This observation might be explained by a faster growth rate and/or higher colonoscopy miss rate for serrated lesions,¹³ which could also underlie the prominent relationship between previous aspirin or EPA treatment and larger serrated-hyperplastic polyps. There was an excess of proximal colorectal polyps during second and subsequent rounds of surveillance of this 'high-risk' cohort of screened individuals, which is consistent with a left-to-right shift reported in other surveillance colonoscopy cohort studies.¹⁴

Increased colorectal adenomatous polyp risk after treatment cessation has previously been reported in three randomised polyp prevention trials of coxib selective cyclooxygenase (COX)-2 inhibitors (APPROVe, PreSAP and APC), which each included an unscheduled 1-2-year off-treatment follow-up period following premature termination of Investigative Medicinal Product use due to cardiovascular safety concerns.¹⁵⁻¹⁷ In the APPROVe and PreSAP trials, the increased risk of any colorectal adenoma in prior coxib users reached

TABLE 4 Interaction between EPA use and clinical factors, baseline EPA status, aspirin use and repeat colonoscopy at baseline during the seAFOod trial.

	IRR (EPA vs no EPA)	p for interaction ^a
Increasing year of age	1.01 (0.98–1.04)	0.65
Sex		
Female	1.20 (0.77–1.85)	0.42
Male	1.04 (0.58–1.86)	
Baseline EPA level (%) ^b		
Low (<0.46%)	0.98 (0.70-1.38)	0.78
High (≥0.46%)	0.91 (0.65–1.28)	
BMI		
Normal (<25 kg/m ²)	1.31 (0.72–2.39)	0.88
Overweight (25–29.9 kg/m²)	1.18 (0.69–2.01)	
Obese (≥30 kg/m²)	1.25 (0.73–2.12)	
Smoking		
Never smoked	0.93 (0.62–1.38)	0.27
Ex-smoker	1.14 (0.79–1.66)	
Current	1.07 (0.60–1.90)	
Units of alcohol per week		
None	1.04 (0.56–1.95)	0.78
1-7	1.17 (0.69–1.96)	
8-21	1.33 (0.79–2.24)	
22+	0.92 (0.51-1.66)	
Aspirin allocation		
Active	0.60 (0.43-0.85)	0.01
Placebo	1.04 (0.75–1.44)	
Repeat colonoscopy in trial		
No	0.89 (0.68–1.15)	0.78
Yes	1.01 (0.62-1.65)	

^aWald test for interaction.

^bRed blood cell % EPA (of total fatty acids) level was dichotomised using the mean baseline EPA level (0.46%).²

statistical significance compared with those previously allocated placebo, with a relative risk of any colorectal adenoma of 1.21 and 1.48, respectively.^{15,16}

However, post-trial colonoscopy outcomes analysis after the Aspirin/Folate Polyp Prevention Study (AFPPS) did not reveal increased colorectal adenoma risk in individuals, who had been randomised to either 81 or 325 mg aspirin for 3 years compared with those allocated placebo and who underwent repeat colonoscopy approximately 4 years later (PDR 39.6% [either aspirin dose] vs. 39.9%; relative risk 1.00 [95% CI 0.80–1.24]).¹⁸ No colorectal polyp number outcomes were reported in the AFPPS follow-up study.¹⁸ The seAFOod trial recruited a 'high risk' cohort with greater colorectal polyp multiplicity at trial entry and post-trial colonoscopic surveillance continued in a quality-assured national programme (that is reflected in the much higher PDR [70%–80%] during follow-up in the BCSP compared with the AFPPS follow-up study),

which could explain the discrepancy in results between the two studies. $^{1,2} \ensuremath{$

Overall, the yield of advanced neoplasia was low, with no CRCs detected during the entire surveillance period. Advanced colorectal polyp detection during post-trial surveillance was similar in individuals who had received aspirin, as opposed to placebo aspirin, during the seAFOod trial. This is consistent with the observation from post-treatment follow-up of the APPROVe trial that prior use of the selective COX-2 inhibitor rofecoxib was associated with increased risk of any colorectal adenoma, but not advanced adenomatous polyps.¹⁵ A valid hypothesis is that advanced colorectal polyps detected during short-term post-trial follow-up represent more established lesions (probably missed at an earlier colonoscopy) that are less amenable to growth repression by aspirin and subsequent de-repression upon treatment cessation.

Our study does not address whether a 'rebound' increase in colorectal polyp risk after cessation of short-term aspirin treatment would manifest as a subsequent increase in CRC incidence. Based on a latent period of at least 8-10 years between randomisation to aspirin and reduced CRC risk (compatible with current understanding of the slow natural history of malignant progression from adenomatous polyps) observed during post-trial follow-up of multiple primary and secondary placebo-controlled vascular prevention trials,^{10,11} the number of individuals followed-up for sufficient time (>20 years) to observe a possible 'rebound' increase in CRC incidence in these studies is limited.^{10,11} Median 10-year observational follow-up of the Women's Health Study (WHS) of 100 mg alternateday aspirin versus placebo, used for 10 years, demonstrated that aspirin use was associated with reduced CRC risk (hazard ratio 0.58 [95% CI 0.42, 0.80]) with a latency of 10 years.¹⁹ However, additional WHS follow-up from 17.5 to 26 years is reported to demonstrate no significant difference in CRC incidence between the group that was originally randomised to aspirin use for 10 years and those allocated placebo (odds ratio 1.16 [95% CI 0.78, 1.72]) suggesting that the aspirin group might have had an excess of CRC events during later follow-up.²⁰ A study of the relationship between the timing or dose of aspirin and CRC risk in the Nurses' Health Study and Health Professionals Follow-Up Study identified a group of individuals with a 'high' duration and dose of prior aspirin use, who had subsequently become 'low' aspirin users (≤5 years or <1.5 tablets per week) during prolonged (>20 years) 'real-life' use of aspirin.²¹ However, this subcohort accounted for only 201 CRC cases and the study did not specifically identify a group that stopped aspirin, highlighting the difficulty of studying CRC risk after aspirin use, even in a large prospective cohort study.²¹

A consistent methodological weakness of post-trial observational follow-up studies, which is shared by our study, is the absence of data on post-trial aspirin usage. At the end of the seAFOod trial, participants were not informed of their treatment allocation and the Plain English Summary of the trial results, which was available online and on request, did not recommend aspirin and/or EPA use. Therefore, one would not expect widespread aspirin use in previous seAFOod trial participants (clinically indicated aspirin use was an exclusion criterion) or that post-trial aspirin use would be unbalanced across the randomised seAFOod trial treatment groups. Moreover, any bias related to post-trial aspirin use would be expected to reduce the 'rebound' effect size compared with the placebo arm based on the reduction in colorectal polyp number associated with current aspirin use.^{1,2}

Another important observation is that combination treatment with aspirin and EPA was associated with significantly lower colorectal polyp recurrence in the seAFOod trial compared with either agent alone. This was apparent in the primary report of the seAFOod trial,^{1,2} but an 'inside the table' analysis of the seAFOod trial data was not pre-specified based on the understanding of the mechanism(s) of action of aspirin and EPA, at the time, which focused on inhibition of COX by both agents, as well as observation that there was no interaction between the two agents when anti-platelet (COX-1-dependent) activity is measured.²² The pharmacological basis of an interaction between aspirin and EPA for colorectal polyp prevention remains unclear. It has been suggested that EPA can be metabolised to a trihydroxy derivative 5,12,18-trihydroxy-EPA (also known as resolving [Rv] E1) by a trans-cellular pathway involving aspirin-acetylated COX-2.⁴ E-type resolvins have inflammation-resolving activity in in vitro and preclinical studies, associated with inhibition of nuclear factor kappa B, which is hypothesised to mediate the additive anti-cancer activity of the combination of aspirin and EPA.²³ RvE1 and its precursor oxylipin 18R-hydroxy-EPA have been detected in multiple different human sample types and clinical disease settings.²⁴ However, RvE1 was not detected (above a limit of detection of 20pg/mL) in plasma and rectal mucosal samples from seAFOod trial participants.²⁵ Although combination of aspirin and EPA treatment were associated with reduced colorectal polyp risk, compared with either agent alone, during the seAFOod trial, we did not observe a larger 'rebound' increase in colorectal polyp risk during post-trial surveillance compared with individuals that had received aspirin alone. The relationship between on-treatment chemoprevention and size of the post-treatment increase in colorectal polyp risk will be dependent on the balance between inhibition of tumour (polyp) initiation versus polyp growth suppression for any given agent(s). This will require further investigation during future polyp prevention trial follow-up and improved understanding of the mechanism(s) of action of aspirin.

This study has several strengths including detailed colorectal polyp outcomes data from a national quality-assured colonoscopy surveillance programme generating a large number of incident colorectal polyps during follow-up. We acknowledge that colorectal polyp nomenclature, as well as risk stratification for colonoscopic surveillance based on colorectal polyp characteristics, has changed during the time encompassed by the seAFOod trial and the post-trial surveillance period, up to the present day.^{7.26} Changes include a clearer distinction between serrated and hyperplastic polyps,²⁶ with the exclusion of diminutive rectal hyperplastic polyps from the BCSP risk stratification algorithm.⁷ However, we continued to describe colorectal polyp outcomes according to the terminology and

reporting tools employed by the BCSP at the time, in order to allow direct comparison between on-trial and post-trial colorectal polyp outcomes.

Study limitations include the lack of data on post-trial aspirin and omega-3 polyunsaturated fatty acid use, absence of data on incident co-morbidities and other drug use, as well as a relatively short post-trial follow-up period, which encompassed only one 'intermediate risk' 3-year colonoscopy for the majority of trial participants. Unfortunately, seAFOod trial participants were not asked to provide consent for future contact at trial entry so we have been unable to contact individuals to enquire about concurrent drug use and permission to access BCSP records at the end of the pre-determined 6-year follow-up period.

Despite the fact that the post-trial surveillance groups, which were defined by randomised trial treatment allocation, were wellmatched for post-trial follow-up duration and the number of surveillance colonoscopies per person, we acknowledge that the complex relationship between colorectal polyp outcomes and the timing and frequency of colonoscopy according to the BCSP surveillance algorithm could introduce bias. However, the secondary analysis was restricted to colonoscopy outcomes at a single time-point 3 years after the seAFOod trial exit colonoscopy confirmed the findings of the main analysis of cumulative colorectal polyp outcomes during follow-up.

We suggest that future biomarker-driven CRC chemoprevention trials include prospective post-intervention follow-up to corroborate our findings that a short-term intervention (aspirin \pm EPA), which is associated with reduced on-treatment colorectal polyp risk, is subsequently linked to 'rebound' increased colorectal polyp incidence when the intervention is stopped. A 'rebound' increase in colorectal polyp risk (and possible increased CRC risk, albeit after a much longer latent period) after short-term aspirin use should be considered as part of the ongoing debate about how best to harness the undoubted cancer-preventative properties of aspirin whilst minimising risk in a precision manner,²⁷ one approach to which is to stop aspirin use in line with age-dependent elevated risk of bleeding.

AUTHOR CONTRIBUTIONS

Amy Downing: Data curation (equal); formal analysis (equal); investigation (equal); methodology (equal); project administration (equal); writing – original draft (equal); writing – review and editing (equal). Hayley Fenton: Data curation (equal); formal analysis (equal); writing – review and editing (equal). Claire Nickerson: Data curation (equal); writing – review and editing (equal). Paul M. Loadman: Funding acquisition (equal); investigation (equal); writing – review and editing (equal). Elizabeth A. Williams: Funding acquisition (equal); investigation (equal); writing – review and editing (equal). Colin J. Rees: Conceptualization (equal); funding acquisition (equal); writing – review and editing (equal). Louise C. Brown: Formal analysis (equal); funding acquisition (equal); methodology (equal); writing – review and editing (equal). Eva J. A. Morris: Conceptualization (equal); data curation (equal); funding

 $\operatorname{AP}_{\!\!\&} \operatorname{T}$ Alimentary Pharmacology & Therapeutics – WILEY

acquisition (equal); investigation (equal); methodology (equal); resources (equal); writing – review and editing (equal). **Mark A. Hull:** Conceptualization (lead); data curation (equal); formal analysis (lead); funding acquisition (lead); investigation (lead); methodology (equal); project administration (lead); resources (equal); supervision (equal); validation (lead); visualization (equal); writing – original draft (lead); writing – review and editing (lead).

ACKNOWLEDGEMENTS

The authors wish to thank all the participants and research staff who contributed to the seAFOod polyp prevention trial. The authors also acknowledge the significant contribution of Mr John Whelpton acting as the patient and public representative in the STOP-ADENOMA Study Group.

Declaration of personal interests: C.J.R. has received grant funding from Medtronic and was an expert witness for Olympus and ARC Medical. All the other authors declare that there are no conflicts of interest.

FUNDING INFORMATION

This project (NIHR128210) was funded by the Efficacy and Mechanism Evaluation (EME) Programme, an MRC and NIHR partnership. The views expressed in this publication are those of the authors and not necessarily those of the MRC, NIHR or the Department of Health and Social Care. MAH and CJR are NIHR Senior Investigators. A.D., E.J.A.M. and M.A.H. are supported by Cancer Research UK grant C23434/A24939.

AUTHORSHIP

Guarantor of the article: Mark A. Hull.

ORCID

Mark A. Hull ^b https://orcid.org/0000-0001-7414-1576

REFERENCES

- Hull MA, Sprange K, Hepburn T, Tan W, Shafayat A, Rees CJ, et al. Eicosapentaenoic acid and aspirin, alone and in combination, for the prevention of colorectal adenomas (seAFOod polyp prevention trial): a multicentre, randomised, double-blind, placebo-controlled, 2×2 factorial trial. Lancet. 2018;392:2583–94.
- Hull MA, Sprange K, Hepburn T, Tan W, Shafayat A, Rees CJ, et al. Eicosapentaenoic acid and/or aspirin for preventing colorectal adenomas during colonoscopic surveillance in the NHS Bowel Cancer Screening Programme: the seAFOod RCT. Efficacy Mech Eval. 2019;6:4–154.
- Cole BF, Logan RF, Halabi S, Benamouzig R, Sandler RS, Grainge MJ, et al. Aspirin for the chemoprevention of colorectal adenomas: meta-analysis of the randomized trials. J Natl Cancer Inst. 2009;101:256–66.
- Oh SF, Pillai PS, Recchiuti A, Yang R, Serhan CN. Pro-resolving actions and stereoselective biosynthesis of 18S E-series resolvins in human leukocytes and murine inflammation. J Clin Invest. 2011;121:569–81.
- Volpato M, Ingram N, Perry SL, Spencer J, Race AD, Marshall C, et al. Cyclooxygenase activity mediates colorectal cancer cell resistance to the omega-3 polyunsaturated fatty acid eicosapentaenoic acid. Cancer Chemother Pharmacol. 2021;87:173–84.

 $WILEY-AP_{\&}T$ Alimentary Pharmacology & Therapeutics

- Wang M, Yu H, Li Z, Gong D, Liu X. Benefits and risks associated with low-dose aspirin use for the primary prevention of cardiovascular disease: a systematic review and meta-analysis of randomized control trials and trial sequential analysis. Am J Cardiovasc Drugs. 2022;22:657–75.
- Rutter MD, East J, Rees CJ, Cripps N, Docherty J, Dolwani S, et al. British Society of Gastroenterology/Association of Coloproctology of Great Britain and Ireland/Public Health England postpolypectomy and post-colorectal cancer resection surveillance guidelines. Gut. 2020;69:201–23.
- Cairns SR, Scholefield JH, Steele RJ, Dunlop MG, Thomas HJ, Evans GD, et al. Guidelines for colorectal cancer screening and surveillance in moderate and high risk groups (update from 2002). Gut. 2010;59:666–89.
- Gupta S, Lieberman D, Anderson JC, Burke CA, Dominitz JA, Kaltenbach T, et al. Recommendations for follow-up after colonoscopy and polypectomy: a consensus update by the US Multi-Society Task Force on colorectal cancer. Gastroenterology. 2020;158:1131–53.
- Flossmann E, Rothwell PM. Effect of aspirin on long-term risk of colorectal cancer: consistent evidence from randomised and observational studies. Lancet. 2007;369:1603–13.
- Rothwell PM, Wilson M, Elwin C-E, Norrving B, Algra A, Warlow CP, et al. Long-term effect of aspirin on colorectal cancer incidence and mortality: 20-year follow-up of five randomised trials. Lancet. 2010;376:1741–50.
- Ngu WS, Bevan R, Tsiamoulos ZP, Bassett P, Hoare Z, Rutter MD, et al. Improved adenoma detection with Endocuff vision: the ADENOMA randomized controlled trial. Gut. 2019;68:280–8.
- Murakami T, Kurosawa T, Fukushima H, Shibuya T, Yao T, Nagahara A. Sessile serrated lesions: clinicopathological characteristics, endoscopic diagnosis, and management. Dig Endosc. 2022;34:1096–109.
- Nusko G, Hahn EG, Mansmann U. Characteristics of metachronous colorectal adenomas found during long-term follow-up: analysis of four subsequent generations of adenoma recurrence. Scand J Gastroenterol. 2009;44:736–44.
- Baron JA, Sandler RS, Bresalier RS, Quan H, Riddell R, Lanas A, et al. A randomized trial of rofecoxib for the chemoprevention of colorectal adenomas. Gastroenterology. 2006;131:1674–82.
- Arber N, Spicak J, Rácz I, Zavoral M, Breazna A, Gerletti P, et al. Five-year analysis of the prevention of colorectal sporadic adenomatous polyps trial. Am J Gastroenterol. 2011;106:1135–46.
- Bertagnolli MM, Eagle CJ, Zauber AG, Redston M, Breazna A, Kim KM, et al. Five-year efficacy and safety analysis of the adenoma prevention with celecoxib trial. Cancer Prev Res. 2009;2:310–21.
- Grau MV, Sandler RS, McKeown-Eyssen G, Bresalier RS, Haile RW, Barry EL, et al. Nonsteroidal anti-inflammatory drug use after 3 years of aspirin use and colorectal adenoma risk: observational follow-up of a randomized study. J Natl Cancer Inst. 2009;101:267-76.

- Cook N, Lee I-M, Zhang S, Moorthy MV, Buring JE. Alternate-day, low-dose aspirin and cancer risk: long-term observational follow-up of a randomized trial. Ann Intern Med. 2013;159:77–85.
- Guirguis-Blake JM, Evans CV, Perdue LA, Bean SI, Senger CA. Aspirin use to prevent cardiovascular disease and colorectal cancer: updated evidence report and systematic review for the US Preventative Services Task Force. JAMA. 2022;327:1585–97.
- 21. Zhang Y, Chan AT, Meyerhardt JA, Giovannucci EL. Timing of aspirin use in colorectal cancer chemoprevention: a prospective cohort study. J Natl Cancer Inst. 2021;113:841–51.
- 22. Larson MK, Ashmore JH, Harris KA. Effects of omega-3 acid ethyl esters and aspirin, alone and in combination, on platelet function in healthy subjects. Thromb Haemost. 2008;100:634–41.
- Ishida T, Yoshida M, Arita M, Nishitani Y, Nishiumi S, Masuda A, et al. Resolvin E1, an endogenous lipid mediator derived from eicosapentaenoic acid, prevents dextran sulfate sodium-induced colitis. Inflamm Bowel Dis. 2010;16:87–95.
- 24. Calder PC. Eicosapentaenoic acid and docosahexaenoic acid derived specialised pro-resolving mediators: concentrations in humans and the effects of age, sex, disease and increased omega-3 fatty acid intake. Biochimie. 2020;178:105-23.
- 25. Fuller H, Race AD, Fenton H, Burke L, Downing A, Williams EA, et al. Plasma and rectal mucosal oxylipin levels during aspirin and eicosapentaenoic acid treatment in the seAFOod polyp prevention trial. Prostaglandins Leukot Essent Fatty Acids. 2023;192:102570. https://doi.org/10.1016/j.plefa.2023.102570
- East JE, Atkin WS, Bateman AC, Clark SK, Dolwani S, Ket SN, et al. British Society of Gastroenterology position statement on serrated polyps in the colon and rectum. Gut. 2017;66:1181–96.
- Drew DA, Chan AT. Aspirin in the prevention of colorectal neoplasia. Ann Rev Med. 2021;72:415–30.

SUPPORTING INFORMATION

Additional supporting information will be found online in the Supporting Information section.

How to cite this article: Downing A, Fenton H, Nickerson C, Loadman PM, Williams EA, Rees CJ, et al. Colorectal polyp outcomes after participation in the seAFOod polyp prevention trial: Evidence of rebound elevated colorectal polyp risk after short-term aspirin use. Aliment Pharmacol Ther. 2023;58:562–572. https://doi.org/10.1111/apt.17646